**Author’s response to reviews**

**Title:** Determinants of deranged thyroid function parameters in children admitted for management of diabetic ketoacidosis/diabetic ketosis

**Authors:**

Peng Shao (shaopeng_2000@163.com)

Shujuan Guo (yemaishuqian@126.com)

Guimei Li (lgmusa2015@163.com)

Daogang Qin (lcsrmyyekqdg@126.com)

Sen Li (lisendr@126.com)

Luan Ying (laurayinger@163.com)

**Version:** 1  **Date:** 08 Apr 2020

**Author’s response to reviews:**

Dear Editors,
Thank you for your letter and suggestion. A few minor revision are list below.
1. Study participants and parents of minors provided signed informed consent at the moment they admitted to the hospital.
2. Some changes in Declarations section.
I hope my article can be accepted finally, of course, I also hope that you can give me more criticism and suggestion.

With my best wishes,
Guimei Li